Author's response to reviews

Title: Possible roles of Epstein-Barr virus in Castleman disease

Authors:

Chih-Hao Chen (musclenet2003@yahoo.com.tw)
Hung-Chang Liu (oncoteam@yahoo.com)
Tzu-Ti Hung (tzdi0921731610@yahoo.com.tw)
Tsang-Pai Liu (liutp@ms1.mmh.org.tw)

Version: 5 Date: 7 July 2009

Author's response to reviews:

Dear Editor

The manuscript had been modified to meet the format, including the format of references and the section of author's contribution. Thanks.

Chih-Hao Chen